Author’s response to reviews

Title: Immune function as predictor of infectious complications and clinical outcome in patients undergoing solid organ transplantation (The ImmuneMo:SOT study): A prospective non-interventional observational trial

Authors:
Camilla Drabe (camilla.h.drabe@gmail.com)
Søren Sørensen (soeren.schwartz.soerensen@regionh.dk)
Allan Rasmussen (allan.rasmussen@regionh.dk)
Michael Perch (michael.perch@regionh.dk)
Finn Gustafsson (finn.gustafsson@regionh.dk)
Omid Rezahosseini (omid.rezahosseini@regionh.dk)
Jens Lundgren (jens.lundgren@regionh.dk)
Sisse Ostrowski (sisse.rye.ostrowski@regionh.dk)
Susanne Dam Nielsen (sdn@dadlnet.dk)

Version: 1 Date: 17 Jun 2019

Author’s response to reviews:

Dear Editor.

Thank you for your comments.

In response:

1) We confirm that informed consent will be obtained after verbal and written information. It is stated in the "Ethics and Consent to Participate" section (Line 350-351).

2) Thank you for noticing that Omid Rezahosseini were not included in the Authors Contributions section. He is now included (Line 388).

3) A clean version with the above mentioned alterations has been uploaded.
Kind Regards,

Camilla Heldbjerg Drabe

On behalf of the study Group.